Author’s response to reviews

Title: Delirium in older hospitalized patients—signs and actions. A retrospective patient record review

Authors:

Yvonne A Johansson (yvonne.a.johansson@vgregion.se)

Ingrid Bergh (ingrid.bergh@his.se)

Iréne Ericsson (irene.ericsson@ju.se)

Elisabeth Kenne Sarenmalm (elisabeth.kenne.sarenmalm@vgregion.se)

Version: 2 Date: 04 Jan 2018

Author’s response to reviews:

Editor Comments:

Authors have made revisions according to reviewers’ comments.

I would suggest that authors remove the figure 1, which is not very informative.

- Authors response:

We now have removed the figure 1 in the manuscript (p.20, L.456).

Additionally, the sentence “Finally, action taken by the healthcare professional is related to patient signs of delirium in a figure (Figure 1)” is removed (p.8, L .182)